Author's response to reviews

Title: Impact of today's media on university student's body image in Pakistan; a conservative, developing country's perspective.

Authors:

Amad N Khan (amadnaseer@gmail.com)
Salema Khalid (inzmania@hotmail.com)
Hussain I Khan (hussain.i.khan@gmail.com)
Mehnaz Jabeen (mehnaz.jabeen@aku.edu)

Version: 8 Date: 22 May 2011

Author's response to reviews: see over
Dear Editor,

We have made the necessary changes required for publication of our article. However, as far as the re-formatting of the tables is concerned, we are not clear as to where and how to proceed and would appreciate it if you could correct this minor revision on our behalf. Thanking you in anticipation.

Best wishes,

Amad N Khan, Salema Khalid, Hussain I Khan, Mehnaz Jabeen.

Dated: 22/05/2011